Author's response to reviews

**Title:** Converting systematic reviews to Cochrane format: a cross-sectional survey of Australian authors of systematic reviews

**Authors:**

Dr Janet H Piehl (jpiehl@bigpond.com)
Sally Green (sally.green@med.monash.edu.au)
Steve McDonald (steve.mcdonald@med.monash.edu.au)

**Version:** 6 **Date:** 14 Jan 2003

We have completed the formatting changes as requested, removing the authors' qualifications from the title page, and reformatting table 2 into the text of the document. We have also altered the language of the text to reflect the discretionary revisions of Dr Handoll. Thank you very much for your assistance.